% Test file for citations (plain bibliography style).
% 
\ifx\eplain\undefined \input eplain \fi
\loggingall

Let's cite chicago: \cite{chicago}.

Now let's have multiple citations: \cite{chicago,texbook}.

Now let's have a non-alphabetic label: \cite{prime-number-theorem}.

Here is an undefined citation: \cite{undefined}.

Here is a citation with an optional note: \cite[note]{texbook}.

Here is a citation with a weird label:
\cite{1234567890)(@!`['";:]|<>./?-=+_$^&}.

Now let's cite an `and others': \cite{others}.

And an accented one: \cite{accented}.

Let's get up to ten refs:
\cite{article-minimal,article-full,article-crossref,whole-journal,%
inbook-minimal,inbook-full}

\let\mbox = \hbox

\bigskip
\leftline{\bf References.}

\bibliography{cite,xampl}
\bibliographystyle{plain}

\end
